Author's response to reviews

Title: Lower estimated glomerular filtration rates in patients on long term lithium; a comparative study and a meta-analysis of literature

Authors:

Chaturaka Rodrigo (chaturaka.rodrigo@gmail.com)
Nipun L de Silva (nipunlak@gmail.com)
Ravindi Gunaratne (ravindi.rr@gmail.com)
Senaka Rajapakse (senaka.ucfm@gmail.com)
Varuni Asanka de Silva (varunidesilva2@yahoo.co.uk)
Raveen Hanwella (raveenhanwella@yahoo.co.uk)

Version: 2 Date: 5 November 2013

Author's response to reviews:

Dear Editor

By mistake we have uploaded an older version of the manuscript. The correct version is uploaded as a revision.

Kind regards
Chaturaka Rodrigo